Author's response to reviews

Title: A Comparison of a New Multinomial Stopping Rule with Stopping Rules of Fleming and Gehan in Single Arm Phase II Cancer Clinical Trials

Authors:

John R Goffin (john.goffin@medportal.ca)
Greg R Pond (gpond@mcmaster.ca)
Dongsheng Tu (dtu@ctg.queensu.ca)

Version: 5 Date: 20 June 2011

Author's response to reviews:

Dear Editors,

As per your request, the following additional changes have been made:

1. Table 5 is now referenced in Results.
2. Authors' contributions is now between competing interests and acknowledgements.
3. Title page has been changed into three sections, with author list, affiliations, and emails, with superscripts added. Telephone and fax numbers were removed following the examples of other articles already published by BMC Medical Research Methodology.

Thank you very much for your guidance.

Sincerely,

John Goffin